Defining phenotypes and treatment effect heterogeneity to inform acute respiratory distress syndrome and sepsis trials: secondary analyses of three RCTs

Manu Shankar-Hari,^{1,2*} Shalini Santhakumaran,³ A Toby Prevost,³ Josie K Ward,⁴ Timothy Marshall,⁴ Claire Bradley,⁴ Carolyn S Calfee,^{5,6,7} Kevin L Delucchi,⁸ Pratik Sinha,⁵ Michael A Matthay,⁵ Jonathan Hackett,⁹ Cliona McDowell,¹⁰ John G Laffey,¹¹ Anthony Gordon,⁴ Cecilia M O'Kane¹² and Daniel F McAuley^{9,10,12}

¹Department of Intensive Care Medicine, Guy's and St Thomas' NHS Foundation Trust, London, UK

 ²School of Immunology and Microbial Sciences, King's College London, London, UK
³Imperial Clinical Trials Unit, School of Public Health, Imperial College London, London, UK

 ⁴Intensive Care Unit, Imperial College/Charing Cross Hospital, London, UK
⁵Department of Medicine, University of California, San Francisco, CA, USA
⁶Department of Anaesthesia, University of California, San Francisco, CA, USA
⁷Cardiovascular Research Institute, University of California, San Francisco, CA, USA
⁸Department of Psychiatry, University of California, San Francisco, CA, USA
⁹Regional Intensive Care Unit, Royal Victoria Hospital, Belfast, UK
¹⁰Northern Ireland Clinical Trials Unit, Royal Victoria Hospital, Belfast, UK
¹¹Anaesthesia, School of Medicine and Regenerative Medicine Institute (REMEDI), CÚRAM Centre for Research in Medical Devices, National University of Ireland Galway, Galway, Ireland

¹²Wellcome-Wolfson Institute for Experimental Medicine, Queen's University Belfast, Belfast, UK

*Corresponding author manu.shankar-hari@kcl.ac.uk

Declared competing interests of authors: Manu Shankar-Hari is funded by a National Institute for Health Research (NIHR) Clinician Scientist Award (reference CS-2016-16-011). A Toby Prevost is a member of the Public Health Research Funding Board. Claire Bradley reports grants from the Wellcome Trust Institution Strategic Support Fund (London, UK) during the conduct of the study. Carolyn S Calfee reports grants from the National Institutes of Health (Bethesda, MD, USA) during the conduct of the study. Carolyn S Calfee also reports grants from GlaxoSmithKline plc (GlaxoSmithKline plc, Brentford, UK), grants and personal fees from Bayer AG (Leverkusen, Germany), personal fees from ProMetic Life Sciences Inc. (Laval, QC, Canada), CSL Behring (King of Prussia, PA, USA), Quark Pharmaceuticals, Inc. (Ness Ziona, Israel) and Roche Holding AG (Basel, Switzerland)/Genentech Inc.

(South San Francisco, CA, USA) outside the submitted work. Michael A Matthay reports grants from National Institutes of Health/National Heart, Lung, and Blood Institute (Bethesda, MD, USA), grants from the US Department of Defense (Arlington, VA, USA), Bayer AG, GlaxoSmithKline plc, and personal fees from Cerus Therapeutics (Concord, CA, USA) outside the submitted work. John G Laffey reports grants from the Health Research Board (Dublin, Ireland) during the conduct of the study. Anthony Gordon reports that he has received speaker fees from Orion Corporation (Espoo, Finland) and Amomed Pharma GmbH (Vienna, Austria). He has consulted for Ferring Pharmaceuticals (Saint-Prex, Switzerland), Tenax Therapeutics, Inc. (Morrisville, NC, USA), Baxter Healthcare (Deerfield, IL, USA), Bristol Myers Squibb (New York, NY, USA) and GlaxoSmithKline plc, and has received non-financial support from Orion Corporation, and grant support from Tenax Therapeutics, Inc. and HCA Healthcare International (London, UK), with funds paid to his institution. He reports grants from the NIHR Efficacy and Mechanism Evaluation (EME) programme, NIHR Research for Patient Benefit programme, NIHR Research Imperial Biomedical Research Centre and from the NIHR Research Professorship award (reference RP-2015-06-018). Cecilia M O'Kane reports grants from the EME programme during the conduct of the study. In addition, she reports personal fees from GlaxoSmithKline plc, C.H. Boehringer Sohn AG & Ko. KG (Ingelheim am Rhein, Germany) and Bayer AG, and grants from the Wellcome Trust and Innovate UK (Swindon, UK) outside the submitted work. Daniel F McAuley reports grants from the NIHR EME programme, Health Research Board, Northern Ireland Public Health Agency Research and Development (Belfast, UK), Intensive Care Society of Ireland and REVIVE for the conduct of this work. Outside the submitted work, Daniel F McAuley reports personal fees from consultancy for GlaxoSmithKline plc, C.H. Boehringer Sohn AG & Ko. KG and Bayer AG. His institution has received funds from grants from NIHR, Wellcome Trust, Innovate UK and others. In addition, Daniel F McAuley is a named inventor on a patent US8962032 covering the use of sialic acid-bearing nanoparticles as anti-inflammatory agents issued to his institution. Daniel F McAuley is a Director of Research for the Intensive Care Society (London, UK) and is the NIHR EME Programme Director. Daniel F McAuley is a member of EME Strategy Advisory Group and a member of the EME Funding Committee.

Published July 2021 DOI: 10.3310/eme08100

Plain English summary

Defining phenotypes and treatment effect heterogeneity Efficacy and Mechanism Evaluation 2021; Vol. 8: No. 10 DOI: 10.3310/eme08100

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

n this project we studied two common conditions that often necessitate admission to an intensive care unit: sepsis and acute respiratory distress syndrome. We found that, although numerous medical treatments are used to treat patients with these two conditions, studies have shown that they have limited success in reducing the risk of dying.

We hypothesised that clinical trials have failed to show benefit because of differences between participants, such that the treatments benefit some patients but harm and/or show no benefit or harm in other patients. To test this theory, we obtained ethics approval to examine, in two separate analyses, clinical and laboratory data from two sepsis trials and one acute respiratory distress syndrome trial.

The first analysis explored whether or not trial participants' risk of dying affected how the treatments worked (referred to as heterogeneity of treatment effect analysis). The treatment effect of the drugs tested in the sepsis trials did not vary with differences in risk of dying, whereas the drug tested in the acute respiratory distress syndrome trial (simvastatin) probably benefited patients with the lowest risk of dying.

The second analysis explored whether or not patients with these conditions can be divided into subgroups in which the treatments have different effects (referred to as latent class analysis). In the case of sepsis, we identified two sepsis subgroups in one trial and three sepsis subgroups in the other trial but found no differences in treatment effect between subgroups in either trial. In the acute respiratory distress syndrome trial we identified two subgroups, and found that treatment was more beneficial in one subgroup.

Our analysis highlights the value of finding participants with greater similarities (subgroups) within sepsis and acute respiratory distress syndrome to help design future clinical trials.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 16/33/01. The contractual start date was in November 2017. The final report began editorial review in July 2019 and was accepted for publication in May 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Shankar-Hari *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk